Author's response to reviews

Title: Changes in practice patterns affecting in-hospital and post-discharge survival among ACS patients

Authors:

Manfred Stommel (stommel@msu.edu)
Ade Olomu (ade.olomu@ht.msu.edu)
Margaret Holmes-Rovner (mholmes@msu.edu)
William Corser (corser@msu.edu)
Joseph C Gardiner (jgardiner@epi.msu.edu)

Version: 5 Date: 23 October 2006

Author's response to reviews: see over
Monday, October 23, 2006

Iratxe Puebla
Senior Assistant Editor
BMC-Series Journal
Journal: BMC Health Services Research

RE: Formatting Changes for MS: 2045310765113416

Dear Ms Puebla,

As requested, we made the following changes in formatting:

(1) Please remove the text 'Stommel et al., Practice Patterns' from the top of each page.
DONE

(2) Keywords section - Please remove the keywords section from the manuscript.
DONE

(3) Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.
DONE

(4) References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.
DONE

(5) Please move the Figure legends to before the tables after the references.
DONE

(6) Please delete the figures that have been included as part of the manuscript and also the accompanying figure placement text. Only the Figure title and description should be part of the manuscript within the 'Figure legends'.

We had some difficulty with this, as the STATA software that produced the figures could not accommodate the information in the accompanying tables (log-rank test results), which were produced in a separate run. We chose to incorporate the main information from the tables into the text, so that there only Figure headings are left in the manuscript.
Sincerely,

Manfred Stommel, PhD
Associate Professor, CON
W-149 Owen Graduate Center
Michigan State University
East Lansing, Michigan 48824
Phone: 517-355-5123
Fax: 517-353-9553
E-mail: stommel@msu.edu